Author’s response to reviews

Title: Systematic review of clinical practice guidelines recommendations about primary cardiovascular disease prevention for older adults.

Authors:

Jesse Jansen (jesse.jansen@sydney.edu.au)
Shannon McKinn (shannon.mckinn@sydney.edu.au)
Carissa Bonner (carissa.bonner@sydney.edu.au)
Les Irwig (les.irwig@sydney.edu.au)
Jenny Doust (jdoust@bond.edu.au)
Paul Glasziou (pglaszio@bond.edu.au)
Brooke Nickel (brooke.nickel@sydney.edu.au)
Barbara van Munster (b.c.vanmunster@amc.uva.nl)
Kirsten McCaffery (kirsten.mccaffery@sydney.edu.au)

Version: 4
Date: 3 July 2015

Author’s response to reviews: see over
Dr Jesse Jansen  
Screening and Test Evaluation Program, Sydney School of Public Health  
July 2, 2015  
Associate Editor  
BMC Family Practice  

Dear Professor Kaduszkiewicz,  


We have addressed your request for additional modifications to the manuscript as detailed below:

• We have removed the Funding section and moved this information to the Acknowledgments.

• We have added a sentence to the Methods section stating that the databases used for our search are publicly available on the internet, and added URLs for all databases.

We have highlighted the changes made in the manuscript.

We extend our thanks for your comments and for considering the revised manuscript for publication in BMC Family Practice.

Sincerely,

Jesse Jansen, on behalf of all authors